Author's response to reviews

Title: Acute kidney injury among adult patients admitted with sepsis in a low income country: Clinical pattern and short term outcomes.

Authors:

Peace Bagasha (bagashap@gmail.com)
Fredrick Nakwagala (nakwagala@yahoo.com)
Arthur Kwizera (kwizera.arthur@gmail.com)
Emmanuel Ssekasanvu (ssekanew@gmail.com)
Robert Kalyesubula (rkalyes@yahoo.com)

Version: 6
Date: 17 November 2014

Author's response to reviews: see over
Dear Dr Bernard Jaar,

Thank you for your very helpful comments. We have attempted to improve the manuscript according to your recommendations, as below;

1. We have corrected the typos through the manuscript.
2. We have harmonized presentation of laboratory values in the text and in tables.
3. The number of AKI cases was 63 but we lost one patient to follow-up. So 62 cases were analysed. We have attempted to clarify that in the textsee end of first paragraph under demographics and first line of paragraph under discharge outcomes.
4. We have incorporated your suggestion for the second paragraph under demographic characteristics in analysis for clinical patterns. Additionally we have reported results in both unadjusted and age adjusted analysis.

We hope that these revisions meet your approval.

Yours faithfully.

Dr Peace Bagasha